

GOPEN ACCESS

Citation: Harris T, Limb ES, Hosking F, Carey I, DeWilde S, Furness C, et al. (2019) Effect of pedometer-based walking interventions on longterm health outcomes: Prospective 4-year followup of two randomised controlled trials using routine primary care data. PLoS Med 16(6): e1002836. https://doi.org/10.1371/journal. pmed.1002836

Academic Editor: Sanjay Basu, Harvard Medical School, UNITED STATES

Received: March 12, 2019

Accepted: May 21, 2019

Published: June 25, 2019

Copyright: © 2019 Harris et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: There are restrictions on the availability of data for this study, due to the initial signed consent agreements around data security, which only allow access to external researchers for research monitoring purposes. Requestors wishing to access the data for the purposes of replicating or checking our analyses can apply to the research data management service at St George's University of London UK (researchdata@sgul.ac.uk), where the data is held RESEARCH ARTICLE

Effect of pedometer-based walking interventions on long-term health outcomes: Prospective 4-year follow-up of two randomised controlled trials using routine primary care data

Tess Harris¹*, Elizabeth S. Limb¹, Fay Hosking¹, Iain Carey¹, Steve DeWilde¹, Cheryl Furness¹, Charlotte Wahlich¹, Shaleen Ahmad¹, Sally Kerry², Peter Whincup¹, Christina Victor³, Michael Ussher^{1,4}, Steve Iliffe⁵, Ulf Ekelund⁶, Julia Fox-Rushby⁷, Judith Ibison⁸, Derek G. Cook¹

Population Health Research Institute, St George's University of London, Tooting, London, United Kingdom,
Pragmatic Clinical Trials Unit, Queen Mary's University of London, London, United Kingdom,
Gerontology and Health Services Research Unit, Brunel University, London, United Kingdom,
Institute for Social Marketing and Public Health, University of Stirling, Stirling, Scotland, United Kingdom,
Research Department of Primary Care and Population Health, University College London, London, United Kingdom,
Department of Sport Medicine, Norwegian School of Sport Sciences, Oslo, Norway,
Department of Propulation Science, King's College London, London, United Kingdom,
Bustitute of Medical and Biomedical Education, St George's University of London, Tooting, London, United Kingdom

* tharris@sgul.ac.uk

Abstract

Background

Data are lacking from physical activity (PA) trials with long-term follow-up of both objectively measured PA levels and robust health outcomes. Two primary care 12-week pedometerbased walking interventions in adults and older adults (PACE-UP and PACE-Lift) found sustained objectively measured PA increases at 3 and 4 years, respectively. We aimed to evaluate trial intervention effects on long-term health outcomes relevant to walking interventions, using routine primary care data.

Methods and findings

Randomisation was from October 2012 to November 2013 for PACE-UP participants from seven general (family) practices and October 2011 to October 2012 for PACE-Lift participants from three practices. We downloaded primary care data, masked to intervention or control status, for 1,001 PACE-UP participants aged 45–75 years, 36% (361) male, and 296 PACE-Lift participants, aged 60–75 years, 46% (138) male, who gave written informed consent, for 4-year periods following randomisation. The following new events were counted for all participants, including those with preexisting diseases (apart from diabetes, for which existing cases were excluded): nonfatal cardiovascular, total cardiovascular (including fatal), incident diabetes, depression, fractures, and falls. Intervention effects on time to first

in a repository. The link is <u>https://doi.org/10.24376/</u> rd.sgul.8138174.

Funding: The research was supported by the National Institute for Health Research (NIHR) UK: PACE-UP trial and 3-year follow-up by the Health Technology Assessment (HTA) Programme NIHR (grant number 10/32/02); PACE-Lift trial by the Research for Patient Benefit (RfPB) Programme NIHR (grant number PB-PG-0909-20055); the PACE-UP 3-year follow-up was also supported by the Collaboration for Leadership in Applied Health Research and Care (CLAHRC) South London. Authors who received funding for the initial trials were TH, DGC, SK, CV, PW, MU, SI, UE, and JF-R.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: TH, ESL, FH, IC, SD, CF, CW, SA, PW, MU, JI, and DGC report grants from NIHR Health Technology Assessment (HTA) Programme, grants from NIHR Research for Patient Benefit Programme (RfPB), and grants from NIHR Collaboration for Leadership in Applied Health Research (CLAHRC) South London, during the conduct of the study for themselves or their institution. SK, CV, SI, UE, and JF-R report grants from NIHR HTA Programme, grants from NIHR RfPB, during the conduct of the study for themselves or their institution. There are no other relationships or activities that could appear to have influenced the submitted work.

Abbreviations: ARR, absolute risk reduction; BMI, body mass index; CI, confidence interval; HADS, hospital anxiety and depression scale; IQR, interquartile range; MET, metabolic equivalent; MVPA, moderate-to-vigorous physical activity; NNT, number needed to treat; NNTH, number needed to harm; PA, physical activity; RCT, randomised controlled trial; REC, Research Ethics Committee. event post-randomisation were modelled using Cox regression for all outcomes, except for falls, which used negative binomial regression to allow for multiple events, adjusting for age, sex, and study. Absolute risk reductions (ARRs) and numbers needed to treat (NNTs) were estimated. Data were downloaded for 1,297 (98%) of 1,321 trial participants. Event rates were low (<20 per group) for outcomes, apart from fractures and falls. Cox hazard ratios for time to first event post-randomisation for interventions versus controls were nonfatal cardiovascular 0.24 (95% confidence interval [CI] 0.07–0.77, p = 0.02), total cardiovascular 0.34 (95% CI 0.12–0.91, p = 0.03), diabetes 0.75 (95% CI 0.42–1.36, p = 0.34), depression 0.98 (95% CI 0.46–2.07, p = 0.96), and fractures 0.56 (95% CI 0.35–0.90, p = 0.02). Negative binomial incident rate ratio for falls was 1.07 (95% CI 0.78–1.46, p = 0.67). ARR and NNT for cardiovascular events were nonfatal 1.7% (95% CI 0.5%-2.1%), NNT = 59 (95% CI 48-194); total 1.6% (95% CI 0.2%–2.2%), NNT = 61 (95% CI 46–472); and for fractures 3.6% (95% CI 0.8%–5.4%), NNT = 28 (95% CI 19–125). Main limitations were that event rates were low and only events recorded in primary care records were counted; however, any underrecording would not have differed by intervention status and so should not have led to bias.

Conclusions

Routine primary care data used to assess long-term trial outcomes demonstrated significantly fewer new cardiovascular events and fractures in intervention participants at 4 years. No statistically significant differences between intervention and control groups were demonstrated for other events. Short-term primary care pedometer-based walking interventions can produce long-term health benefits and should be more widely used to help address the public health inactivity challenge.

Trial registrations

PACE-UP isrctn.com ISRCTN98538934; PACE-Lift isrctn.com ISRCTN42122561.

Author summary

Why was this study done?

- Health benefits from physical activity interventions require sustained increases in physical activity levels, yet evidence of long-term objective increases in physical activity and effects on health outcomes is lacking.
- Routine data from primary care records can provide robust health outcome measures but have been little used in evaluating physical activity interventions.

What did the researchers do and find?

• Two primary care 12-week pedometer-based walking interventions in 45- to 75-yearolds increased objectively measured moderate-to-vigorous physical activity levels by one-third between baseline and 3–4 years and were associated with significant decreases in both cardiovascular events (heart attacks, strokes, etc.) and fractures at 4 years.

• Approximately 60 people needed to receive the walking intervention to prevent one cardiovascular event and approximately 28 people to prevent one fracture.

What do these findings mean?

- Short-term pedometer-based walking interventions can help adults and older adults to achieve not only sustained increases in physical activity but also important long-term health benefits.
- Routine primary care data provide an efficient way to provide long-term trial follow-up for assessing health outcomes.

Introduction

Strong evidence exists that physical activity (PA) is protective for a wide range of health conditions [1–3], and inactivity is claimed to be the fourth leading risk factor for global mortality [2]. Meta-analyses of cohort studies have reported clear benefits of moderate-intensity PA for many chronic diseases, including diabetes [4], ischaemic heart disease [4], stroke [4], fractures [5], and depression [6]. However, all these estimates are based on evidence from observational cohort studies, in which individual baseline differences in PA levels, assessed by questionnaires, were linked to subsequent disease outcomes. PA questionnaires are known to be inaccurate and subject to recall bias [7], and there is the possibility of regression dilution bias in such studies, which would lead to underestimating benefits. This raises the question of whether changes in PA that occur after PA interventions will have similar or possibly larger effects. An additional advantage of trial data is that changes in PA have usually been objectively measured.

Many PA interventions, including pedometer-based walking interventions, have shown short-term increases in PA levels [8–10], but to achieve the long-term health benefits demonstrated above in cohort studies, increases in PA need to be sustained, and long-term trial data with objectively measured PA outcomes and robust health outcomes are limited, with calls for more such trials [10-12]. The Finnish Diabetes Prevention Study observed significant PA increases and showed impressive 58% reductions in type 2 diabetes, but it combined dietary and PA interventions; therefore, it is difficult to estimate independent PA effects [13]. In contrast, a recent large primary care trial failed to reduce type 2 diabetes incidence, despite increasing PA levels [14]. Trials examining PA effects on cardiovascular outcomes have shown mixed results: some showing strong protective effects on both heart attacks [15,16] and strokes [16] but others failing to reduce cardiovascular events [17]. Primary care trials that have successfully increased PA levels have shown both increased [18] and reduced [19] self-reported falls, but a systematic review of exercise interventions in older adults demonstrated a significant reduction in fall-related fractures [20]. A meta-analysis of PA interventions demonstrated a reduction in depressive symptoms [21]. However, only one of the trials above used routinely recorded primary care or hospital data to capture events [15]; for other trials, participants were asked about health events, and these were validated by checking patient records, leading to important potential reporting bias. The benefit of using routine electronic records for extended follow-up of trials has been established in a different context by the West of Scotland Coronary Prevention Study, which showed that routine records gave very similar results to rigorously collected clinical trial data for cardiovascular events and deaths [22]. Using routine records in trials has the additional merit that if participants have given permission to access their health records, these may be available even when participants are lost to follow-up.

We conducted two pedometer-based walking trials with adults and older adults (PACE-UP, PACE-Lift), which increased accelerometer-measured step count and moderate-to-vigorous PA (MVPA) levels in bouts at 12 months [23,24] and with sustained PA increases at 3–4 years [25]. Across both trials at 3–4 years, all intervention groups were doing approximately an extra 30 minutes per week of MVPA compared with baseline, up about a third on their baseline levels [25]. Both trials recruited through primary care, and participant consent to link trial data with primary care record data was sought. The aim of the present paper was to evaluate the intervention effects from PACE-UP and PACE-Lift on longer-term health outcomes relevant to the walking interventions, using routinely collected primary care data.

Methods

Study design and participants

Two primary care randomised controlled trials (RCTs) of effective 12-week pedometer-based walking interventions are included, both providing long-term health outcome data from routine primary care records: PACE-UP, which recruited 45- to 75-year-olds from seven London (United Kingdom) practices from October 2012 to November 2013, and PACE-Lift, which recruited 60- to 75-year-olds from three Berkshire and Oxfordshire (UK) practices from October 2011 to October 2012. The trials were similar in their primary care recruitment and in the 12-week pedometer-based walking interventions, incorporating behaviour-change techniques [26, 27]. Participants were very similar in terms of their baseline characteristics, apart from deprivation level (see Table 1), but randomisation ensured even distribution of deprivation levels across intervention and control groups [23,24], and deprivation was not an effect modifier [23]. At long-term follow-up, 3-year findings for both PACE-UP intervention groups (postal and nurse-supported) and 4-year findings from the PACE-Lift (nurse-supported) intervention group all showed very similar effects on PA levels [25]. Given the similarities of these trials and their sustained PA effects, we therefore present a combined analysis of all three intervention groups at 4 years on primary care outcomes. All analyses adjust for study as a covariate.

PACE-UP trial. Trial methods are published [27], as are 3- and 12-month [23] and 3-year [25] findings, cost-effectiveness analyses [28], and funder's report [29]. All research participants gave written informed consent, and permission was sought for researchers to access data from their primary care records. Ethical approval was granted by London Hampstead Research Ethics Committee (REC) (UK) (12L/LO/0219), including substantial amendments for extended follow-up work. The trial had a control arm (usual care) and two intervention arms: postal, who received the 12-week PACE-UP walking programme (pedometer, handbook, and PA diary) by post, and nurse-supported, who received the same materials at the first of three practice nurse PA consultations. Baseline findings for participants included in primary care outcomes analyses are summarised in Table 1, and the postal and nurse interventions are summarised in Table 2. The handbook and diary are available at www.paceup.sgul.ac.uk/ materials. The protocol, approved by ethics before extended follow-up commenced, included details of long-term follow-up, primary care data download procedures, and primary care data

	PACE-UP study (<i>N</i> = 1,001)	PACE-Lift study $(N = 297)^a$
	n (%)	n (%)
Age at randomisation		
45-59 years	520 (52%)	0 (0%)
60–75 years	481 (48%)	297 (100%)
Gender: male	361 (36%)	138 (46%)
Marital status: married	645 (66%)	240 (81%)
National quintiles of Index of Multiple Deprivation r	ank	
1–3 (most deprived)	553 (57%)	29 (10%)
4	212 (22%)	51 (17%)
5 (least deprived)	201 (21%)	217 (73%)
Ethnicity		
White	776 (80%)	289 (99%)
Asian/Asian British	68 (7%)	2 (1%)
Black/African/Caribbean/Black British	96 (10%)	1 (0%)
Other	24 (2%)	1 (0%)
Current smoker	80 (8%)	16 (6%)
General health ^b : Very good or good	802 (82%)	259 (89%)
Self-reported pain ^b	674 (69%)	201 (69%)
Limiting long-standing illness	216 (22%)	72 (26%)
Townsend disability score ^b		
None (0)	586 (59%)	205 (70%)
Slight or some disability (1–6)	371 (38%)	83 (28%)
Appreciable or severe disability (7–18)	29 (3%)	6 (2%)
HADS depression score ^b : borderline or high	108 (11%)	
Geriatric depression score ^b : high		19 (7%)
Overweight/obese : BMI ≥ 25 kg/m ²	666 (67%)	200 (67%)
Preexisting disease on GP records		
Cardiovascular disease	64 (6%)	31 (10%)
Diabetes	70 (7%)	18 (6%)
Depression	99 (10%)	34 (11%)
Accelerometry data		
Average adjusted baseline step count per day		
Mean (sd)	7,492 (2,675)	7,331 (2,829)
Median (IQR)	7,344 (5,567–9,106)	7,043 (5,302–9,124)
Total weekly minutes of MVPA in \geq 10-minute bouts		
Mean (sd)	94 (102)	92 (109)
Median (IOR)	65 (21 to 133)	53 (3 to 140)

Table 1. Baseline characteristics of the PACE-UP and PACE-Lift cohorts.

^aOne PACE-Lift participant in the control group died before 12 months and is included in this table. The participant is not included in the analysis of primary care record data but is included in the fatal + nonfatal cardiovascular events analysis.

^bFull references for general health, self-reported pain, Townsend disability score, HADS score, and geriatric depression score are given in the trial protocols [26,27].

Abbreviations: BMI, body mass index; GP, general practitioner; HADS, hospital anxiety and depression scale; IQR, interquartile range; MVPA, moderate to severe physical activity.

https://doi.org/10.1371/journal.pmed.1002836.t001

Table 2. Components of interventions for PACE-UP and PACE-Lift trials^a.

Component	PAG	PACE-Lift		
	Postal	Nurse	Nurse	
Pedometer Yamax Digi-Walker (Tokyo, Japan) SW-200. Provides step count, requires daily manual recording and resetting	Posted with instructions for use ^b	Given with instructions by nurse at first appointment	Given with instructions by nurse at first appointment	
Dedicated practice nurse PA	Not applicable	3 consultations	4 consultations,	
consultations (including		Week 1 "First Steps" (approx. 30 mins)	Week 1 "First Steps" (approx. 45 mins)	
benaviour-change techniques)		Week 5 "Continuing the Changes" (approx. 20 mins)	Week 3 "Continuing the Changes" (approx. 30 mins)	
		Week 9 "Building Lasting Habits" (approx. 20 mins)	Week 7 "Keeping up the Changes" (approx. 30 mins)	
			Week 11 "Building Lasting Habits" (approx. 30 mins)	
Accelerometer feedback as part of intervention	Not applicable	Not applicable	Actigraph GT3X+ (accelerometer) worn for 1 week prior to each nurse appointment. Nurse downloaded accelerometer data during consultation and provided immediate feedback on time spent in sedentary, light, moderate, and vigorous PA levels in relation to activities recorded in PA diary.	
Handbook (including behaviour- change techniques)	Posted ^b	Given by nurse at first appointment	Given by nurse at first appointment	
Target-setting: step count goals and PA goals and use of walking planner	Blinded pedometer (Yamax DigiWalker CW200) worn for 7 days at baseline to calculate average daily baseline steps, used to set step count targets. Use of 12-week walking planner. Advised to add 1,500 steps/day and then 3,000 steps/day to average baseline steps in graded manner over 12 weeks.	Blinded pedometer (Yamax DigiWalker CW200) worn for 7 days at baseline to calculate average daily baseline steps, used to set step count targets. Use of 12-week walking planner. Advised to add 1,500 steps/day and then 3,000 steps/day to average baseline steps in a graded manner over 12 weeks.	Nurses discussed appropriate step count and PA goals with participants based on baseline step count and weekly time in MVPA from accelerometry and any health issues. Participants encouraged to set both step count and time in MVPA goals, encouraged to start low and go slow. Walking planner to help them plan when and where and with whom they planned to walk. Goals reviewed and reset at each	
	"3,000-steps-in-30-minutes" message for PA intensity.	Targets could be adapted in discussion with nurse.		
		"3,000-steps-in-30-minutes" message for PA intensity.		
12-week PA and step count diary (including behaviour-change techniques)	Posted ^b and encouraged to return completed diary to researchers after 12-week intervention.	Given by nurse at first appointment, reviewed by nurse at other appointments and encouraged to return completed diary to researchers after 12-week intervention.	Given by nurse at first appointment and reviewed at each nurse appointment.	

^aThis table has been adapted from tables in the published trial protocols [26,27]. These are open-access articles published under licence to BioMed Central and distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction, provided the work is properly cited (http://creativecommones.org/licenses/by/2.0).

^bResearcher telephoned 1 week later to check whether programme had arrived.

Abbreviations: approx., approximately; MVPA, moderate-to-vigorous PA; PA, physical activity.

https://doi.org/10.1371/journal.pmed.1002836.t002

outcomes and is included (<u>S1 Text</u>), as is a CONSORT checklist for primary care data followup (<u>S2 Text</u>).

PACE-Lift trial. Trial methods are published [26], as are 3- and 12-month [24] and 4-year [25] findings. Written informed consent was gained from all participants, and permission was sought for researchers to access data from participants' primary care records. Initial ethical approval was granted by Oxfordshire REC C (UK) (11/H0606/2) up to 12-month follow-up; the study was then closed and new ethical approval from this REC was granted for

4-year follow-up (15/SC0352), which required us to reconsent participants at 4 year to access further data from their primary care records, as trial follow-up was being restarted. The trial had a control arm (usual care) and an intervention arm whose participants received the 12-week PACE-Lift walking programme (pedometer, handbook, PA diary, and feedback on their accelerometry measures) at the first of four practice nurse PA consultations. Baseline findings for participants included in primary care outcome analyses are summarised in Table 1, and the intervention is summarised in Table 2. The protocol, approved by ethics before extended follow-up commenced, included details of long-term follow-up, primary care data download, and primary care data outcomes and is included (S3 Text), as is a CONSORT checklist for primary care data follow-up (S4 Text).

Sample size

Power calculations for each of the trials have been described previously in the PACE-UP [27] and PACE-Lift [26] trial protocols and were related to the primary outcome of change in objectively measured PA levels at prespecified time points. The sample size was determined by the number of trial participants providing consent for primary care data download, and the confidence intervals (CIs) around our estimates give a clear indication of the level of precision of our findings.

Outcome measures

A priori we wanted to evaluate intervention effects on long-term health outcomes relevant to walking interventions assessed from primary care records 4 years post-baseline. These outcome measures were not described in the original published trial protocols [26, 27], as they were not planned at the time of study design, but they were prespecified in the protocols approved by ethics (S1 and S3 Text), ahead of long-term data collection, and were designed to be measurable retrospectively from primary care data. We defined these health outcomes as nonfatal cardiovascular events (myocardial infarction, coronary artery bypass graft, angioplasty, transient ischaemic attack, and stroke), total cardiovascular events (cardiovascular deaths plus nonfatal cardiovascular events), new onset diabetes, new onset depression episode, injurious falls (those recorded in primary care records), and fractures. Cardiovascular disease and depression outcomes were estimated both for those with and without prior cardiovascular disease or depression, respectively. We also examined the effect of the interventions on number of primary care consultations (excluding the 3-month intervention period, as consultations were part of the nurse intervention arms for both trials).

Procedures

We downloaded primary care data for trial participants who gave written consent, minus those who subsequently withdrew from the trials, for the 4-year periods following randomisation from the seven PACE-UP and three PACE-Lift practices. Data were downloaded at 12 months (at end of initial trial follow-up) and at 4 years (after extended follow-up) at all 10 practices. If participants had both sets of data, the 12-month data were not needed, as they were duplicated in the 4-year data. For those without any data at 4 years, 12-month data were used, if available. See CONSORT diagram (Fig 1) for details of numbers of participants at each time point with primary care data for both trials. Data on participants were censored if a patient left the practice or if they died whilst still registered at the practice. Searches were set up to download the following information from primary care records: Read codes for diseases (including those arising from hospital admissions) and consultation data (see S1 Fig for details of exactly how events were counted). For cardiovascular events and depression, we separated



Fig 1. (a) PACE-UP and (b) PACE-Lift CONSORT diagrams for primary care records analyses. Complete data for 4 years were only available on those participants who were registered at the same primary care practice at baseline and 4 years. 7 deaths in PACE-UP participants (2 Control. 5 Intervention) and 4 deaths in PACE-Lift participants (2 Control. 2 Intervention) were recorded on registered patients. Data for those participants who moved away or had died before 4 years are censored when they left the practice or at their date of death.

https://doi.org/10.1371/journal.pmed.1002836.g001

events occurring in participants with and without preexisting disease, so Read codes for these diseases that occurred prior to the individual randomisation date were taken as evidence of preexisting disease. Prior Read codes for diabetes were used to exclude those with preexisting diabetes at date of randomisation from subsequent diabetes analyses only, so that we could estimate numbers of new type 2 diabetes diagnoses. Data were processed blind to trial group, by a researcher without access to trial data apart from randomisation dates, according to an agreed protocol. Two medically qualified researchers, also blind to intervention group, separately checked that the events counted were appropriate. Details on deaths (including date and cause) were collected systematically for both trials from general practices prior to recontacting participants for long-term follow-up (at 3 years for PACE-UP and 4 years for PACE-Lift) and were therefore available for all participants for this period of follow-up (regardless of whether they had given permission for their primary care records to be downloaded), apart from any patients who had deregistered with their practice by moving away. Known deaths from cardiovascular causes were counted as outcome events and included in total cardiovascular events; other deaths were treated as censored data. Once all events had been verified, the primary care data and trial data were linked.

Statistical analysis

Analyses were carried out in STATA version 15.0 (StataCorp). We combined data from both trials and modelled the effect of the interventions on cardiovascular, diabetes, fracture, and depression outcomes using Cox regression models to estimate Cox hazard ratios (and 95% CIs) for time to first event post-randomisation, adjusting for age (as a continuous variable), sex, and study. For cardiovascular disease and depression diagnosis. For all these outcomes, we also calculated Kaplan-Meier survival curves. For falls and consultations, in which we wanted to allow for multiple events to be counted, we used negative binomial regression, which models the counts as a Poisson process but allows for overdispersion. These models were used to estimate incident rate ratios (and 95% CIs), also adjusting for age, sex, and study. We estimated absolute risk reductions (ARRs) and used the approach recommended by Altman to calculate the number needed to treat (NNT) for each estimate and its confidence limits [30, 31].

Patient and public involvement

Pilot work with older primary care patients from three general practices was carried out prefunding, with focus groups discussing ideas for a pedometer-based PA intervention. They provided input into study design—for example, encouraging postal recruitment and recruitment of couples as well as individuals. Both trials had a patient advisor as a Trial Steering Committee member; they were involved in discussions about study conduct and advised on patient materials, dissemination of results to participants, and safety reporting. All trial participants were provided with individual feedback after 12-month follow-up. Trial results were disseminated at the following times: after baseline assessments, after analysis of the main 12-month results, and after 3-year (PACE-UP) or 4-year (PACE-Lift) follow-up. A trial website summarising trial results and publications was created for PACE-UP (http://www.paceup.sgul.ac.uk) and circulated to participants. Intervention burden was assessed by nurse group participants as part of trial process evaluations [32] and by samples of all the intervention groups as part of the qualitative evaluations [33, 34].

Results

Overall, 98% (1,297/1,321) of initial trial participants gave written consent for primary care data linkage and had their data downloaded—1,001/1,023 (98%) from PACE-UP and 296/298 (99%) from PACE-Lift at baseline—and 223/225 (99%) of PACE-Lift participants recontacted at 4-year follow-up. Primary care data available at different time points are shown in the two flow diagrams (Fig 1a and 1b). Overall, 82% (1,077/1,321) of participants had 4 years of complete data: 85% (871/1,023) of PACE-UP and 69% (206/298) of PACE-Lift participants.

Table 3 presents the time to first event for each outcome; the model coefficients are given in S1 Table. For nonfatal cardiovascular events, in both trials the proportion of events was lower in the intervention than in the control group, both for those without a prior cardiovascular diagnosis and for all participants. The hazard ratios for all participants were 0.24 (95% CI 0.07–0.77, p = 0.02) (demonstrated in a Kaplan-Meier plot in Fig 2) and for those without a prior diagnosis 0.27 (95% CI 0.08–0.88, p = 0.03). When fatal cardiovascular events were included, results were similar: hazard ratios of 0.34 (95% CI 0.12–0.91, p = 0.03) for all participants and 0.31 (95% CI 0.11–0.93, p = 0.04) for those without a prior diagnosis. In terms of new diabetes diagnoses, there was no statistically significant intervention effect, with a hazard ratio of 0.75 (95% CI 0.42–1.36, p = 0.34). Similarly, for new depression diagnoses, there was no overall effect of the intervention, hazard ratio 0.98 (95% CI 0.46–2.07, p = 0.96) in all participants and 0.92 (95% CI 0.41–2.03, p = 0.83) in those without a prior diagnosis. For fractures,

	PACE-UP			PACE-Lift ^a			PACE-UP and PACE-Lift			HRs ^b	
Event Recorded in Primary Care Record	N	Control (<i>n</i> = 333)	Intervention (<i>n</i> = 668)	N	Control (<i>n</i> = 147)	Intervention (<i>n</i> = 149)	N	Control (<i>n</i> = 480)	Intervention (n = 817)	HR (95% CI)	p-Value
Nonfatal cardiovascular events											
All participants	1,001	7 (2.1%)	3 (0.4%)	296	3 (2.0%)	1 (0.7%)	1,297	10 (2.1%)	4 (0.5%)	0.24 (0.07-0.77)	0.02
No previous cardiac diagnosis	937	6 (1.9%)	3 (0.5%)	265	3 (2.3%)	1 (0.8%)	1,202	9 (2.0%)	4 (0.5%)	0.27 (0.08–0.88)	0.03
Nonfatal and fatal cardiov	ascular ev	vents									
All participants	1,001	7 (2.1%)	4 (0.6%)	297	4 (2.7%)	2 (1.3%)	1,298	11 (2.3%)	6 (0.7%)	0.34 (0.12-0.91)	0.03
No previous cardiac diagnosis	937	6 (1.9%)	3 (0.5%)	266	4 (3.0%)	2 (1.5%)	1,203	10 (2.3%)	5 (0.7%)	0.31 (0.11–0.93)	0.04
Diabetes diagnosis											
No previous diabetes diagnosis	931	16 (5.2%)	21 (3.4%)	278	4 (2.9%)	5 (3.5%)	1,209	20 (4.5%)	26 (3.4%)	0.75 (0.42–1.36)	0.34
Depression diagnosis			·								
All participants	1,001	11 (3.3%)	15 (2.2%)	296	0 (0.0%)	4 (2.7%)	1,297	11 (2.3%)	19 (2.3%)	0.98 (0.46-2.07)	0.96
No previous depression diagnosis	902	10 (3.3%)	13 (2.2%)	262	0 (0.0%)	3 (2.3%)	1,164	10 (2.3%)	16 (2.2%)	0.92 (0.41–2.03)	0.83
Fractures											
All participants	1,001	28 (8.4%)	26 (3.9%)	296	8 (5.4%)	8 (5.4%)	1,297	36 (7.5%)	34 (4.2%)	0.56 (0.35-0.90)	0.02

Table 3. PACE-UP and PACE-Lift studies. Hazard ratios for first events after randomisation in primary care records.

^aOne PACE-Lift participant in the intervention group died before 12 months, and primary care records are not available. This participant has only been included in the analyses of nonfatal + fatal cardiovascular events, so the denominators for these analyses are 150 for the PACE-Lift intervention group and 818 for the combined intervention group.

^bEstimates from the Cox regression models are adjusted for age, sex, and study.

Abbreviation: HR, hazard ratio.

https://doi.org/10.1371/journal.pmed.1002836.t003

in PACE-UP the proportion of patients with a fracture during follow-up was lower in the intervention group (26/668, 3.9%) compared with the control group (28/333, 8.4%); for PACE-Lift both groups had similar proportions of fractures, 5.4% (8/149) in the intervention group and 5.4% (8/147) in the control group. The overall Cox regression hazard ratio across both trials was significantly reduced (0.56, 95% CI 0.35–0.90, p = 0.02). Fig 2 also shows these findings in a Kaplan-Meier time-to-first-event diagram.

Table 4 shows the findings for medically reported falls and consultations; S1 Table shows the coefficients. Overall, approximately 14% (115/818) of intervention and 13% (63/480) of control participants had one or more falls during follow-up; the overall incident rate ratio for falls was 1.07 (95% CI 0.78–1.46, p = 0.67). The mean number of consultations per year over 4-year follow-up was similar in intervention and control groups for both trials (approximately 6, with a standard deviation of 5), and the incident rate ratio was 1.01 (95% CI 0.93–1.10, p = 0.82), showing no intervention effect on number of consultations.

Table 5 shows the ARRs and NNTs (95% CIs) for each event, combined across both trials. The results for cardiovascular events and fractures (for which our interventions were associated with a protective effect) for all participants were as follows: nonfatal cardiovascular events ARR 1.7% (0.5%–2.1%), NNT 59 (48–194); total cardiovascular events ARR 1.6% (0.2%–2.2%), NNT 61 (46–472); fractures ARR 3.6% (0.8%–5.4%), NNT 28 (19–125).

Discussion

Statement of principal findings

Ninety-eight percent of trial participants gave written consent for their primary care records to be used and had data successfully downloaded. Our results indicate that 12-week pedometer-



subsequent implementation studies.

Strengths and weaknesses of the study

Fig 2. Kaplan-Meier curves for time to first event post-randomisation recorded in primary care records.



b) Diabetes diagnosis



when participants knew their PA levels were being measured. Our study thus demonstrates the advantage of using routine data to evaluate long-term health outcomes in trials and also in

Important strengths were that both trials recruited from primary care and that we sought participant consent to use their primary care data, thus allowing us to examine long-term health



	PAG	CE-UP	PACE-Lift		PACE-UP and PACE-Lift			
	Control Intervention		Control Intervention		Control Intervention		Incident Rate Ratio ^a	
	(<i>n</i> = 333)	(<i>n</i> = 668)	(n = 147)	(<i>n</i> = 149)	(<i>n</i> = 480)	(<i>n</i> = 818)	IRR (95% CI)	<i>p</i> -Value
Number of falls								
	n (%)	n (%)	n (%)	n (%)	n (%)	n (%)		
0	285 (86%)	574 (86%)	132 (90%)	128 (86%)	417 (87%)	702 (86%)		
1	39 (12%)	80 (12%)	10 (7%)	15 (10%)	49 (10%)	95 (12%)		
2	8 (2%)	11 (2%)	2 (1%)	3 (2%)	10 (2%)	14 (2%)		
3+	1 (0.3%)	3 (0.3%)	3 (2%)	3 (2%)	4 (1%)	6 (1%)		
Falls and consultation	ons: Rate per yea	r of follow-up						
	Mean (sd)	Mean (sd)	Mean (sd)	Mean (sd)	Mean (sd)	Mean (sd)		
Falls	0.04 (0.12)	0.05 (0.14)	0.04 (0.14)	0.05 (0.15)	0.04 (0.13)	0.05 (0.14)	1.07 (0.78-1.46)	0.67
Consultations ^b	5.9 (4.8)	6.1 (5.1)	6.1 (4.6)	6.2 (5.9)	5.9 (4.8)	6.1 (5.3)	1.01 (0.93–1.10)	0.82

Table 4. PACE-UP and PACE-Lift studies. Incident rate ratios for falls and consultations after randomisation from primary care records.

^aNegative binomial models were used to estimate incident rate ratios. These model the count as Poisson with extra variation; the dispersion parameter is the expected mean. All models adjust for age, sex, and study.

^bConsultations in the first 90 days after randomisation are not included, as intervention group participants in both studies had practice nurse consultations during this time as part of the intervention.

https://doi.org/10.1371/journal.pmed.1002836.t004

outcomes using routinely collected data. Having two trials with similar recruitment methods and interventions, which we had previously combined for long-term follow-up [25], meant that we increased the power of our analyses. A key strength of UK routinely recorded primary care data, which may not apply to other healthcare systems, is that it fully captures secondary care diagnoses from accident and emergency, outpatients, and hospital admissions (National Health Service and private), in addition to primary care consultations and diagnoses, thus

Table 5. PACE-UP and PACE-Lift studies combined: ARRs and NNTs for p	primary care recorded events.
--	-------------------------------

Event recorded in primary care record	N	ARR ^a (95% CI)	NNT ^b (95% CI)
Nonfatal cardiovascular events			
All participants	1,297	1.7 (0.5–2.1)	59 (48–194)
No previous cardiac diagnosis	1,202	1.6 (0.3–2.0)	62 (50–386)
Nonfatal and fatal cardiovascular events			
All participants	1,298	1.6 (0.2–2.2)	61 (46–472)
No previous cardiac diagnosis	1,203	1.7 (0.2–2.2)	60 (46-562)
Diabetes diagnosis			
No previous diabetes diagnosis	1,209	1.2 (-1.7 to 2.8)	84 (NNTH 59 to ∞ to NNTB 35)
Depression diagnosis			
All participants	1,297	0.1 (-2.7 to 1.4)	1,873 (NNTH 37 to ∞ to NNTB 72)
No previous depression diagnosis	1,164	0.2 (-2.6 to 1.5)	463 (NNTH 39 to ∞ to NNTB 66)
Fractures			
All participants	1,297	3.6 (0.8-5.4)	28 (19–125)

^aARR is calculated as 1/NNT.

^bNNT/NNTB is the number needed to treat to show benefit from the intervention (i.e., prevent one event) at 4 years. Where the 95% CI for ARR is consistent with an increase in risk from the intervention, the NNTH is also shown [31]. NNT is calculated using the formula 1/{SurvC**HR-SurvC} where SurvC is the Kaplan-Meier survival probability in the control group at 4 years and HR is the hazard ratio [30].

Abbreviations: ARR, absolute risk reduction; NNT, number needed to treat; NNTH, number needed to harm.

https://doi.org/10.1371/journal.pmed.1002836.t005

reducing potential for bias in outcome assessment. The inevitable losses that occur when collecting long-term trial follow-up data can be reduced by using routinely collected data for outcome assessment. We followed up 67% (681/1,023) of participants at 3 years in PACE-UP [25] and 76% (225/298) at 4 years for PACE-Lift with objective PA primary outcome data, but for these analyses using routinely collected primary care data at 4 years, we had 98% (1,297/1,321) contributing to analyses and 82% (1,077/1,321) providing complete data over the whole 4 years, despite needing to reconsent PACE-Lift participants at 4-year follow-up. Others have shown how higher levels of electronic follow-up (93%) compared with fieldwork follow-up (83%) at 2 years retained more of those in the most deprived groups [35], demonstrating reduced potential for selection bias.

An important limitation is that we did not have complete 4 years of follow-up data for all participants; however, differences were already evident in cardiovascular events and fractures by 12-month follow-up, when we had over 97% complete primary care data across all groups. We were also constrained by the data routinely recorded in primary care records, which do not reflect all cases. For cardiovascular events and fractures, underrecording is unlikely to have been a problem, as major diagnoses like myocardial infarction, stroke, coronary artery bypass graft, fractures, etc., are well recorded in primary care records following hospital notification of events. For falls and new depression episodes, cases occurring in the community but not reported to primary care will be missed, and for diabetes, new cases are only diagnosed when blood tests are done. However, although cases will be underrecorded in primary care records for these conditions, there is no reason for this to differ by intervention status, particularly as the intervention did not affect consultation rate, so this should not have led to bias. A further limitation is the uncertainty of our estimates. Given the low event rate in this sample and the relatively short follow-up period for events, the CIs are wide, indicating uncertainty regarding the exact magnitudes of effect.

Comparisons with previous studies

Details of all the intervention studies that are referred to in this section, including more information on the specific interventions and their effects on PA levels, are summarised in <u>S2 Table</u>.

Cardiovascular events. The reductions we demonstrated in nonfatal (0.24 [95% CI 0.07-0.77]) and total (0.34 [95% CI 0.12-0.91]) cardiovascular events are consistent with the effect in older primary care patients 2 years after a 9-month PA programme that significantly increased self-reported PA and significantly decreased blood pressure and lipids (reduction in cardiovascular events risk ratio 0.15 [95% CI 0.04-0.51]) [15]. Similarly, others showed significant reductions in both heart attacks (relative risk 0.51) and strokes (relative risk 0.52) alongside self-reported PA increases, 6 months post-intervention in community-based hypertension patients [16]. However, Newman et al found no cardiovascular event reduction in the PA group at 2.6-year follow-up in older adults with functional limitations (hazard ratio 1.10 [95% CI 0.85-1.42]) [17]. Possible reasons were that the PA intervention group had more opportunity to report events; cardiovascular disease levels were high, possibly precipitating events or reducing potential benefits; and there was a possible suboptimal activity dose, as their moderate PA cutoff was >760 counts/minute [17], much lower than ours (\geq 1,952 counts/minute) [26, 27].

Comparisons with risk estimates from cohort studies are more difficult. A key issue is that all cohort studies in systematic reviews are based on questionnaire PA measures, with their known inaccuracies and recall bias [7]; the variety of different questionnaires also makes it difficult to standardise how PA is quantified. Cohort studies' effect estimates usually compare inactive participants with those achieving much higher PA levels. Thus, a recent good-quality systematic review provides relative risks for an 11.25 metabolic equivalent (MET) hour/week increase in PA levels compared with being inactive of 0.69 (95% CI 0.67–0.71) [36], corresponding to international recommendations of 150 minutes of MVPA weekly in \geq 10-minute bouts [37]. Our intervention increased PA by about 30 minutes of MVPA in bouts weekly long-term, one-fifth of 150 minutes, so after scaling down the relative risks from that paper [36] (S5 Text), their cardiovascular incidence effect estimate for the same level of PA increase becomes 0.98 (95% CI 0.97–0.99), suggesting a much more modest effect on cardiovascular events than from our intervention study.

Type 2 diabetes. We showed a nonsignificant reduction in type 2 diabetes cases (0.75 [95% CI 0.42–1.36]), consistent with a pedometer intervention in UK primary care prediabetes patients, which increased PA by 498 steps/day (162–834) across the 3-year follow-up and achieved a hazard ratio for type 2 diabetes cases of 0.74 (95% CI 0.48–1.14) [14]. The significant 58% reduction in diabetes incidence seen in the Finnish Diabetes Prevention trial (with a combined PA/dietary intervention) occurred with not only an increase in PA levels (86% of the intervention group achieving >4 hour/week of exercise compared with 71% of controls, p = 0.001) but also dramatic weight loss (4.2 ± sd 5.1 kg in the intervention group versus 0.8 ± sd 3.7 kg in controls) [13].

From cohort studies, the scaled-down estimate of effect for type 2 diabetes incidence (as done for cardiovascular disease, above) to compare appropriately with the PA increase achieved in our trials was 0.93 (95% CI 0.92–0.93) (S5 Text), consistent with our estimate.

Fractures. The reduction in fractures observed in the intervention groups compared with controls 0.56 (95% CI 0.35–0.90) is consistent with findings from a systematic review of exercise interventions in older adults that gave a pooled risk ratio of 0.60 (95% CI 0.45–0.85) [20] for fall-related fractures. However, only one of the 15 included studies was a pure walking intervention, and several studies focused on strength and balance training and did not include any walking component [20].

A meta-analysis of 13 cohort studies reported a relative risk for fractures of 0.62 (95% CI 0.56–0.69) in women [5] and 0.55 (95% CI 0.44–0.69) in men [5]. As for cardiovascular disease, these benefits are based on varying PA levels at baseline, but generally considerably greater than 30 minutes of MVPA from walking weekly, and thus need scaling back. Given the heterogeneity of exposure difference in the meta-analysis, it is difficult to provide such a scaled-back estimate.

Falls. We did not observe any reduction in medically reported falls (overall negative binomial regression incident rate ratio 1.07 [95% CI 0.78–1.46]). However, only primary care– recorded falls were captured. Although others have seen falls reduced by exercise interventions 0.86 (95% CI 0.77–0.94) [20] and 0.77 (95% CI 0.71–0.83) [38], most interventions included balance, resistance, and strength training and not simply walking. Walking is something of a paradox for risk of falling, as it can increase falls by increasing exposure. Those interventions that have reduced falls successfully in older adults have usually additionally included balance components [19, 20, 38], whereas pure walking interventions have sometimes increased falls [18], though not always [39].

A cohort study examining the association between regular walking and falls amongst community-dwelling older adults found that it did not increase falls for those at low risk (hazard ratio 0.88 [95% CI 0.48–1.62]), but it significantly increased risk in those with two or more risk factors for falling (hazard ratio 1.89 [95% CI 1.04–3.43]) [40], suggesting that the relationship between walking and falls is complex.

Depression. We found no intervention effect on new depression cases (hazard ratio 0.98 [95% CI 0.46–2.07]), although CIs were very wide. A systematic review and meta-analysis of PA interventions reported reduced depression symptoms (standardised mean effect size 0.37 [95% CI 0.24–0.50] for supervised PA studies and 0.52 [95% CI 0.28–0.77] for unsupervised

studies) [21], supported by a systematic review of exercise referral schemes, which also showed reduced clinical depression risk in the two studies reporting this outcome (pooled standard-ised mean difference -0.82 [95% CI -1.28 to -0.35]) [41]. Of note, most studies included in the reviews reported depression risk based on symptom scores rather than clinical diagnoses of depression; we extracted the latter from primary care records, which could underestimate depression risk, as it relies on patients presenting with symptoms and general (family) practitioners detecting and recording them. Our negative finding from primary care records is, however, consistent with our main trial outcomes, which showed no intervention effects on depressive symptoms at 12 months [23, 24] or 3–4 years [25].

Cohort studies support an association between PA levels and depression, with increased PA levels associated with reduced depression risk, with the majority of studies (25/30) in a systematic review reporting this, though no meta-analyses or forest plots were presented [42], so there is no overall effect estimate for direct comparison with our findings.

Implications for clinicians, researchers, and policy makers

An important implication for both future clinical practice and policy is that primary care short-term pedometer-based walking interventions incorporating behaviour-change techniques can lead not only to long-term changes in PA levels but also to long-term beneficial health effects for adults and older adults. They could thus help to address the public health physical inactivity challenge and be part of the 'call to activity' for clinicians and patients [43]. This supports current guidance to promote pedometers alongside support for goal-setting, self-monitoring, and feedback [44] and suggests that policy makers should consider investment in this short-term primary care pedometer-based walking intervention because of its proven long-term health benefits. Our previous work has shown that sustained effects on PA levels were similar for postal and nurse-supported intervention groups [25] and that the postal route was more cost-effective [28]; this route therefore seems most promising to pursue for implementation.

Our demonstration of the widespread acceptance by trial participants for their primary care records to be accessed and the feasibility of using these data for evaluating long-term health outcomes is consistent with others' findings [22, 35]. Our experience supports initiatives from funders such as the Medical Research Council and the National Institute for Health Research in the UK and other funders internationally, to encourage researchers undertaking trials to include options for longer-term data collection from routine records in their funding applications. We found only one other example of a PA trial that used routine primary care data to assess long-term health outcomes [15]. Using such outcomes provides objective evidence applying to all those randomised (not just those who complete trial follow-up), and by focusing on the clinical benefits, it also avoids the problems inherent in measuring change in PA levels in large numbers of subjects. Linking healthcare data to trial data can strengthen the evaluation and implementation of primary care–based interventions and should be more widely explored by researchers.

Unanswered questions and future research

Observational studies strongly suggest that the dose-response association between PA levels and both all-cause mortality [45] and a range of chronic diseases [36] is nonlinear, with the greatest benefit appearing when changing from a sedentary lifestyle to low levels of activity and smaller additional benefits from higher levels of activity [36, 45]. Such data have influenced the newly published Physical Activity Guidelines for Americans [46]. Our data, based on RCT evidence, are important for silencing the sceptics who argue that the new guidelines

lack high-quality evidence from RCTs [43]. It is too soon to conclude with any confidence that relatively modest changes in PA can achieve greater health benefits than predicted from cohort studies, but that is the hope. Observational studies, even those using objective PA measures, cannot provide direct evidence of what happens when individuals change their PA levels by modest amounts; what is needed is for more PA trials that have successfully increased PA levels to provide long-term follow-up of clinical outcomes.

Conclusions

We showed significant reductions in both cardiovascular events and fractures for the PA intervention groups from two trials over a 4-year period, supporting and extending the benefits demonstrated by the long-term trial PA outcomes. However, our CIs are wide, indicating uncertainty regarding the exact magnitudes of the effects. We also confirmed the feasibility of using routine primary care data to provide robust long-term health outcomes for RCTs. Shortterm 12-week pedometer-based walking interventions can have long-term positive health effects and should be used more widely to help address the public health physical inactivity challenge.

Supporting information

S1 Fig. Details of how events were counted in downloaded primary care data. (TIF)

S1 Text. Protocol for extended follow-up and implementation research for PACE-UP trial including primary care data download. (DOC)

S2 Text. CONSORT checklist for the PACE-UP trial 4-year primary care data outcomes paper.

(DOCX)

S3 Text. Protocol for extended follow-up of PACE-Lift trial including primary care data download.

(DOC)

S4 Text. CONSORT checklist for the PACE-Lift trial 4-year primary care data outcomes paper.

(DOCX)

S5 Text. Comparison of physical activity levels from our study with cohort studies for estimating cardiovascular and diabetes relative risks. (DOCX)

S1 Table. Coefficients for Cox regression and negative binomial models. (DOCX)

S2 Table. Details of physical activity intervention studies with health outcomes. (DOCX)

Acknowledgments

We remember and thank our colleague and coinvestigator, Dr. Sunil Shah, who died in September 2015. We would like to thank the Oxfordshire and Berkshire (UK) PACE-Lift general practices, their practice nurses who supported this study, and all the patients who participated: Sonning Common Health Centre, Oxfordshire; Balmore Park Surgery, Reading, Berkshire; and Priory Avenue Surgery, Reading, Berkshire. Similarly, we wish to thank all the South-West London (UK) PACE-UP general practices, their practice nurses who supported this study, and all the patients from these practices who participated: Upper Tooting Road Practice, Tooting; Chatfield Practice, Battersea; Wrythe Green Practice, Carshalton; Francis Grove Practice, Wimbledon; Putneymead Practice, Putney; Heathfield Practice, Putney; and Cricket Green Practice, Mitcham. We would also like to thank both of our supportive Trial Steering Committees.

The views and opinions expressed herein are those of the authors and not necessarily those of the National Institute for Health Research, the National Health Service, or the Department of Health and Social Care.

Author Contributions

Conceptualization: Tess Harris, Elizabeth S. Limb, Fay Hosking, Iain Carey, Steve DeWilde, Shaleen Ahmad, Sally Kerry, Peter Whincup, Christina Victor, Michael Ussher, Steve Iliffe, Ulf Ekelund, Julia Fox-Rushby, Judith Ibison, Derek G. Cook.

Data curation: Tess Harris, Steve DeWilde, Cheryl Furness, Charlotte Wahlich.

Formal analysis: Elizabeth S. Limb, Fay Hosking, Iain Carey, Derek G. Cook.

Funding acquisition: Tess Harris, Sally Kerry, Peter Whincup, Christina Victor, Michael Ussher, Steve Iliffe, Ulf Ekelund, Julia Fox-Rushby, Derek G. Cook.

Methodology: Tess Harris.

Project administration: Cheryl Furness.

Supervision: Tess Harris.

Validation: Tess Harris, Elizabeth S. Limb, Derek G. Cook.

- Writing original draft: Tess Harris, Elizabeth S. Limb, Fay Hosking, Iain Carey, Derek G. Cook.
- Writing review & editing: Tess Harris, Elizabeth S. Limb, Fay Hosking, Iain Carey, Steve DeWilde, Cheryl Furness, Charlotte Wahlich, Shaleen Ahmad, Sally Kerry, Peter Whincup, Christina Victor, Michael Ussher, Steve Iliffe, Ulf Ekelund, Julia Fox-Rushby, Judith Ibison, Derek G. Cook.

References

- Department of Health. Start Active, Stay Active: A report on physical activity for health from the four home countries' Chief Medical Officers. 2011. <u>https://www.gov.uk/government/publications/start-</u> active-stay-active-a-report-on-physical-activity-from-the-four-home-countries-chief-medical-officers
- Lee IM, Shiroma EJ, Lobelo F, Puska P, Blair SN, Katzmarzyk PT. Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. Lancet. 2012; 380(9838):219–29. https://doi.org/10.1016/S0140-6736(12)61031-9 PMID: 22818936
- Piercy KL, Troiano RP, Ballard RM, Carlson SA, Fulton JE, Galuska DA, et al. The Physical Activity Guidelines for Americans. JAMA. 2018; 320(19):2020–8. https://doi.org/10.1001/jama.2018.14854 PMID: 30418471.
- Kyu HH, Bachman VF, Alexander LT, Mumford JE, Afshin A, Estep K, et al. Physical activity and risk of breast cancer, colon cancer, diabetes, ischemic heart disease, and ischemic stroke events: systematic review and dose-response meta-analysis for the Global Burden of Disease Study 2013. BMJ. 2016; 354:i3857. https://doi.org/10.1136/bmj.i3857 PMID: 27510511.

- Moayyeri A. The association between physical activity and osteoporotic fractures: a review of the evidence and implications for future research. Ann Epidemiol. 2008; 18(11):827–35. https://doi.org/10.1016/j.annepidem.2008.08.007 PMID: 18809340.
- Zhai L, Zhang Y, Zhang D. Sedentary behaviour and the risk of depression: a meta-analysis. Br J Sports Med. 2015; 49(11):705–9. https://doi.org/10.1136/bjsports-2014-093613 PMID: 25183627.
- Tudor-Locke CE, Myers AM. Challenges and opportunities for measuring physical activity in sedentary adults. Sports Med. 2001; 31(2):91–100. https://doi.org/10.2165/00007256-200131020-00002 PMID: 11227981
- Bravata DM, Smith-Spangler C, Sundaram V, Gienger AL, Lin N, Lewis R, et al. Using pedometers to increase physical activity and improve health: a systematic review. JAMA. 2007; 298(19):2296–304. https://doi.org/10.1001/jama.298.19.2296 PMID: 18029834
- Kang M, Marshall SJ, Barreira TV, Lee JO. Effect of pedometer-based physical activity interventions: a meta-analysis. Res Q Exerc Sport. 2009; 80(3):648–55. https://doi.org/10.1080/02701367.2009. 10599604 PMID: <u>19791652</u>
- Hobbs N, Godfrey A, Lara J, Errington L, Meyer TD, Rochester L, et al. Are behavioral interventions effective in increasing physical activity at 12 to 36 months in adults aged 55 to 70 years? A systematic review and meta-analysis. BMC Med. 2013; 11:75. <u>https://doi.org/10.1186/1741-7015-11-75</u> PMID: 23506544
- Richards J, Thorogood M, Hillsdon M, Foster C. Face-to-face versus remote and web 2.0 interventions for promoting physical activity. Cochrane Database Syst Rev. 2013; 9:CD010393. https://doi.org/10. 1002/14651858.CD010393.pub2 PMID: 24085593
- National Institute for Health and Care Excellence. Behaviour change: individual approaches. 2014. https://www.nice.org.uk/guidance/ph49/resources/behaviour-change-individual-approaches-pdf-1996366337989.
- Tuomilehto J, Lindstrom J, Eriksson JG, Valle TT, Hamalainen H, Ilanne-Parikka P, et al. Prevention of type 2 diabetes mellitus by changes in lifestyle among subjects with impaired glucose tolerance. N Engl J Med. 2001; 344(18):1343–50. https://doi.org/10.1056/NEJM200105033441801 PMID: 11333990.
- Davies MJ, Gray LJ, Troughton J, Gray A, Tuomilehto J, Farooqi A, et al. A community based primary prevention programme for type 2 diabetes integrating identification and lifestyle intervention for prevention: the Let's Prevent Diabetes cluster randomised controlled trial. Prev Med. 2016; 84:48–56. https:// doi.org/10.1016/j.ypmed.2015.12.012 PMID: 26740346.
- Arija V, Villalobos F, Pedret R, Vinuesa A, Timon M, Basora T, et al. Effectiveness of a physical activity program on cardiovascular disease risk in adult primary health-care users: the "Pas-a-Pas" community intervention trial. BMC Public Health. 2017; 17(1):576. <u>https://doi.org/10.1186/s12889-017-4485-3</u> PMID: 28619115.
- Gong J, Chen X, Li S. Efficacy of a Community-Based Physical Activity Program KM2H2 for Stroke and Heart Attack Prevention among Senior Hypertensive Patients: A Cluster Randomized Controlled Phase-II Trial. PLoS ONE. 2015; 10(10):e0139442. https://doi.org/10.1371/journal.pone.0139442 PMID: 26426421.
- Newman AB, Dodson JA, Church TS, Buford TW, Fielding RA, Kritchevsky S, et al. Cardiovascular Events in a Physical Activity Intervention Compared With a Successful Aging Intervention: The LIFE Study Randomized Trial. JAMA Cardiol. 2016; 1(5):568–74. https://doi.org/10.1001/jamacardio.2016. 1324 PMID: 27439082.
- Lawton BA, Rose SB, Elley CR, Dowell AC, Fenton A, Moyes SA. Exercise on prescription for women aged 40–74 recruited through primary care: two year randomised controlled trial. BMJ. 2008; 337: a2509. https://doi.org/10.1136/bmj.a2509 PMID: 19074218
- Iliffe S, Kendrick D, Morris R, Griffin M, Haworth D, Carpenter H, et al. Promoting physical activity in older people in general practice: ProAct65+ cluster randomised controlled trial. Br J Gen Pract. 2015; 65(640):e731–8. https://doi.org/10.3399/bjgp15X687361 PMID: 26500320.
- **20.** Zhao R, Feng F, Wang X. Exercise interventions and prevention of fall-related fractures in older people: a meta-analysis of randomized controlled trials. Int J Epidemiol. 2017; 46(1):149–61. https://doi.org/10. 1093/ije/dyw142 PMID: 27477031.
- Conn VS. Depressive symptom outcomes of physical activity interventions: meta-analysis findings. Ann Behav Med. 2010; 39(2):128–38. https://doi.org/10.1007/s12160-010-9172-x PMID: 20422333.
- Barry SJ, Dinnett E, Kean S, Gaw A, Ford I. Are routinely collected NHS administrative records suitable for endpoint identification in clinical trials? Evidence from the West of Scotland Coronary Prevention Study. PLoS ONE. 2013; 8(9):e75379. https://doi.org/10.1371/journal.pone.0075379 PMID: 24058681.
- 23. Harris T, Kerry SM, Limb ES, Victor CR, Iliffe S, Ussher M, et al. Effect of a Primary Care Walking Intervention with and without Nurse Support on Physical Activity Levels in 45- to 75-Year-Olds: The

Pedometer And Consultation Evaluation (PACE-UP) Cluster Randomised Clinical Trial. PLoS Med. 2017; 14(1):e1002210. https://doi.org/10.1371/journal.pmed.1002210 PMID: 28045890.

- Harris T, Kerry SM, Victor CR, Ekelund U, Woodcock A, Iliffe S, et al. A primary care nurse-delivered walking intervention in older adults: PACE (pedometer accelerometer consultation evaluation)-Lift cluster randomised controlled trial. PLoS Med. 2015; 12(2):e1001783. <u>https://doi.org/10.1371/journal.pmed.1001783</u> PMID: 25689364
- 25. Harris T, Kerry SM, Limb ES, Furness C, Wahlich C, Victor CR, et al. Physical activity levels in adults and older adults 3–4 years after pedometer-based walking interventions: Long-term follow-up of participants from two randomised controlled trials in UK primary care. PLoS Med. 2018; 15(3):e1002526. https://doi.org/10.1371/journal.pmed.1002526 PMID: 29522529.
- Harris T, Kerry S, Victor C, Ekelund U, Woodcock A, Iliffe S, et al. Randomised controlled trial of a complex intervention by primary care nurses to increase walking in patients aged 60–74 years: protocol of the PACE-Lift (Pedometer Accelerometer Consultation Evaluation—Lift) trial. Bmc Public Health. 2013; 13. https://doi.org/10.1186/1471-2458-13-5 PMID: 23289648
- Harris T, Kerry SM, Victor CR, Shah SM, Iliffe S, Ussher M, et al. PACE-UP (Pedometer and consultation evaluation—UP)—a pedometer-based walking intervention with and without practice nurse support in primary care patients aged 45–75 years: study protocol for a randomised controlled trial. Trials. 2013; 14:418. https://doi.org/10.1186/1745-6215-14-418 PMID: 24304838
- Anokye N, Fox-Rushby J, Sanghera S, Cook DG, Limb E, Furness C, et al. Short-term and long-term cost-effectiveness of a pedometer-based exercise intervention in primary care: a within-trial analysis and beyond-trial modelling. BMJ Open. 2018; 8(10):e021978. https://doi.org/10.1136/bmjopen-2018-021978 PMID: 30337309.
- Harris T, Kerry S, Victor C, Iliffe S, Ussher M, Fox-Rushby J, et al. A pedometer-based walking intervention in 45- to 75-year-olds, with and without practice nurse support: the PACE-UP three-arm cluster RCT. Health Technol Assess. 2018; 22(37):1–274. https://doi.org/10.3310/hta22370 PMID: 29961442.
- Altman DG, Andersen PK. Calculating the number needed to treat for trials where the outcome is time to an event. BMJ. 1999; 319(7223):1492–5. https://doi.org/10.1136/bmj.319.7223.1492 PMID: 10582940.
- **31.** Altman DG. Confidence intervals for the number needed to treat. BMJ. 1998; 317(7168):1309–12. https://doi.org/10.1136/bmj.317.7168.1309 PMID: 9804726.
- Furness C, Howard E, Limb E, Cook DG, Kerry S, Wahlich C, et al. Relating process evaluation measures to complex intervention outcomes: findings from the PACE-UP primary care pedometerbased walking trial. Trials. 2018; 19(1):58. <u>https://doi.org/10.1186/s13063-017-2428-z</u> PMID: 29357921.
- Victor CR, Rogers A, Woodcock A, Beighton C, Cook DG, Kerry SM, et al. What factors support older people to increase their physical activity levels? An exploratory analysis of the experiences of PACE-Lift trial participants. Arch Gerontol Geriatr. 2016; 67:1–6. <u>https://doi.org/10.1016/j.archger.2016.06.006</u> PMID: 27394028
- 34. Normansell R, Smith J, Victor C, Cook DG, Kerry S, Iliffe S, et al. Numbers are not the whole story: a qualitative exploration of barriers and facilitators to increased physical activity in a primary care based walking intervention. Bmc Public Health. 2014; 14.
- Davies G, Jordan S, Brooks CJ, Thayer D, Storey M, Morgan G, et al. Long term extension of a randomised controlled trial of probiotics using electronic health records. Sci Rep. 2018; 8(1):7668. <u>https://doi.org/10.1038/s41598-018-25954-z PMID: 29769554</u>.
- **36.** Wahid A, Manek N, Nichols M, Kelly P, Foster C, Webster P, et al. Quantifying the Association Between Physical Activity and Cardiovascular Disease and Diabetes: A Systematic Review and Meta-Analysis. J Am Heart Assoc. 2016; 5(9). https://doi.org/10.1161/JAHA.115.002495 PMID: 27628572.
- World Health Organisation. Global Recommendations on Physical Activity for Health. 2010. <u>https://www.who.int/dietphysicalactivity/publications/9789241599979/en/</u>
- Picorelli AM, Pereira LS, Pereira DS, Felicio D, Sherrington C. Adherence to exercise programs for older people is influenced by program characteristics and personal factors: a systematic review. J Physiother. 2014; 60(3):151–6. https://doi.org/10.1016/j.jphys.2014.06.012 PMID: 25092418.
- Okubo Y, Osuka Y, Jung S, Rafael F, Tsujimoto T, Aiba T, et al. Walking can be more effective than balance training in fall prevention among community-dwelling older adults. Geriatr Gerontol Int. 2016; 16(1):118–25. https://doi.org/10.1111/ggi.12444 PMID: 25613322.
- 40. Okubo Y, Seino S, Yabushita N, Osuka Y, Jung S, Nemoto M, et al. Longitudinal association between habitual walking and fall occurrences among community-dwelling older adults: analyzing the different risks of falling. Arch Gerontol Geriatr. 2015; 60(1):45–51. https://doi.org/10.1016/j.archger.2014.10.008 PMID: 25456885.

- **41.** Pavey TG, Taylor AH, Fox KR, Hillsdon M, Anokye N, Campbell JL, et al. Effect of exercise referral schemes in primary care on physical activity and improving health outcomes: systematic review and meta-analysis. BMJ. 2011; 343:d6462. https://doi.org/10.1136/bmj.d6462 PMID: 22058134
- Mammen G, Faulkner G. Physical activity and the prevention of depression: a systematic review of prospective studies. Am J Prev Med. 2013; 45(5):649–57. https://doi.org/10.1016/j.amepre.2013.08.001 PMID: 24139780.
- Thompson PD, Eijsvogels TMH. New Physical Activity Guidelines: A Call to Activity for Clinicians and Patients. JAMA. 2018; 320(19):1983–4. https://doi.org/10.1001/jama.2018.16070 PMID: 30418469.
- 44. National Institute for Health and Care Excellence. Walking and cycling. Local measures to promote walking and cycling as forms of travel or recreation. 2012. Report No.: NICE Public Health Guidance 41.
- Woodcock J, Franco OH, Orsini N, Roberts I. Non-vigorous physical activity and all-cause mortality: systematic review and meta-analysis of cohort studies. Int J Epidemiol. 2011; 40(1):121–38. <u>https://doi.org/10.1093/ije/dyq104</u> PMID: 20630992.
- US Department of Health and Human Service. Physical Activity Guidelines for Americans. 2nd ed. Washington DC: US Dept of Health and Human Services; 2018.